Non-invasive tests for clinically significant portal hypertension after HCV cure

Georg Semmler, M.D, Sabela Lens, M.D, Elias L. Meyer, M.Sc, Anna Baiges, M.D, Edilmar Alvarado-Tapias, M.D, Elba Llop, M.D, Luis Tellez, M.D, Philipp Schwabl, M.D, Ezequiel Mauro, M.D, Laia Escudé, M.D, Cristina Díez, M.D, Luis Ibañez-Samaniego, M.D, Ángela Puente, M.D, José Ignacio Fortea, M.D, Marta Abadía, M.D, Alberto Zanetto, M.D, Andrés Conthe, M.D, Helena Hernandez-Évole, M.D, Irina Sofia Luzko Scheid, M.D, Jidong Jia, M.D., Ph.D, Hitoshi Yoshiji, M.D., Ph.D, Sven M. Francque, M.D., Ph.D, Emmanuel A. Tsochatzis, M.D, Francesco Paolo Russo, M.D., Ph.D, Gonzalo Crespo, M.D, Xavier Forns, M.D, Rafael Bañares, M.D, Càndid Villanueva, M.D, Virginia Hernández-Gea, M.D., Ph.D, Thomas Reiberger, M.D, Jaume Bosch, M.D, Juan Carlos García-Pagán, M.D, Mattias Mandorfer, M.D., Ph.D., A study by the Baveno Cooperation: an EASL consortium

PII: S0168-8278(22)03056-2

DOI: https://doi.org/10.1016/j.jhep.2022.08.025

Reference: JHEPAT 8860

To appear in: Journal of Hepatology

Received Date: 25 January 2022

Revised Date: 26 July 2022

Accepted Date: 11 August 2022

Please cite this article as: Semmler G, Lens S, Meyer EL, Baiges A, Alvarado-Tapias E, Llop E, Tellez L, Schwabl P, Mauro E, Escudé L, Díez C, Ibañez-Samaniego L, Puente Á, Ignacio Fortea J, Abadía M, Zanetto A, Conthe A, Hernandez-Évole H, Sofia Luzko Scheid I, Jia J, Yoshiji H, Francque SM, Tsochatzis EA, Paolo Russo F, Crespo G, Forns X, Bañares R, Villanueva C, Hernández-Gea V, Reiberger T, Bosch Juan J, García-Pagán C, Mandorfer M, A study by the Baveno Cooperation: an EASL consortium, Non-invasive tests for clinically significant portal hypertension after HCV cure, *Journal of Hepatology* (2022), doi: https://doi.org/10.1016/j.jhep.2022.08.025.

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published

in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

© 2022 European Association for the Study of the Liver. Published by Elsevier B.V. All rights reserved.



Non-invasive tests for clinically significant portal hypertension after HCV cure

Georg SEMMLER^{1,2,*}, M.D.

Sabela LENS^{3,4,5,*}, M.D.

Elias L. MEYER⁶, M.Sc.

- Anna BAIGES^{3,4,5}, M.D.
- Edilmar ALVARADO-TAPIAS^{5,7}, M.D.
- Elba LLOP^{5,8}, M.D.
- Luis TELLEZ⁹, M.D.
- Philipp SCHWABL^{1,2,10}, M.D.
- Ezequiel MAURO¹¹, M.D.
- Laia ESCUDÉ^{3,4,5}, M.D.
- Cristina DÍEZ^{12,13}, M.D.
- Luis IBAÑEZ-SAMANIEGO^{5,12,14}, M.D.
- Ángela PUENTE¹⁵, M.D.
- José Ignacio FORTEA¹⁵, M.D.
- Marta ABADÍA¹⁶, M.D.
- Alberto ZANETTO¹⁷, M.D.
- Andrés CONTHE^{5,14}, M.D.
- Helena HERNANDEZ-ÉVOLE^{3,4}, M.D.
- Irina Sofia LUZKO SCHEID^{3,4}, M.D.
- Jidong JIA^{18,19,20}, M.D., Ph.D.
- Hitoshi YOSHIJI²¹, M.D., Ph.D.
- Sven M. FRANCQUE^{22,23,24,25}, M.D., Ph.D.
- Emmanuel A. TSOCHATZIS^{10,26}, M.D.
- Francesco Paolo RUSSO¹⁶, M.D., Ph.D.

Gonzalo CRESPO^{3,4,5}, M.D. Xavier FORNS^{3,4,5}, M.D. Rafael BAÑARES^{12,13,14,27}, M.D. Càndid VILLANUEVA^{5,7}, M.D. Virginia HERNÁNDEZ-GEA^{3,4,5}, M.D., Ph.D. Thomas REIBERGER^{1,2}, M.D. Jaume BOSCH^{3,4,5,28}, M.D. Juan Carlos GARCÍA-PAGÁN^{3,4,5,**}, M.D. Mattias MANDORFER^{1,2,**}, M.D., Ph.D. A study by the Baveno Cooperation: an EASL consortium

*G.S. and S.L. contributed equally to the work.

**J.C.G-P. and M.M. share the corresponding and last author position.

¹ Division of Gastroenterology and Hepatology, Department of Internal Medicine III, Medical University of Vienna, Vienna, Austria

² Vienna Hepatic Hemodynamic Lab, Division of Gastroenterology and Hepatology, Department of Internal Medicine III, Medical University of Vienna, Vienna, Austria
³ Liver Unit, Hospital Clínic, Universitat de Barcelona, Health Care Provider of the European Reference Network on Rare Liver Disorders (ERN-Liver), Barcelona, Spain
⁴ August Pi i Sunyer Biomedical Research Institute (IDIBAPS), Universitat de Barcelona, Barcelona, Spain
⁵ Centro de Investigación Biomédica En Red de Enfermedades Hepáticas y Digestivas (CIBERehd), Instituto de Salud Carlos III, Spain

⁶ Institute for Medical Statistics, Center for Medical Statistics, Informatics and Intelligent Systems, Medical University Vienna, Vienna, Austria

⁷ Hospital of Santa Creu and Sant Pau, Autonomous University of Barcelona, Hospital Sant Pau Biomedical Research Institute (IIB Sant Pau), Barcelona, Spain

⁸ Liver Unit, Hospital Universitario Puerta De Hierro Majadahonda, Universidad Autònoma de Madrid, Madrid, Spain

⁹ Department of Gastroenterology and Hepatology, Hospital Universitario Ramón y Cajal, IRYCIS, University of Alcalá, Madrid, Spain

¹⁰ UCL Institute for Liver and Digestive Health, Royal Free Hospital and UCL, London, United Kingdom

¹¹ Liver Unit and Liver Transplant Unit, Hospital Italiano, Buenos Aires, Argentina

¹² Unidad de Enfermedades Infecciosas/VIH, Hospital General Universitario Gregorio Marañón, Madrid, Spain

¹³ Instituto de Investigación Sanitaria Gregorio Marañón (IiSGM), Madrid, Spain

¹⁴ Liver Unit, Hospital General Universitario Gregorio Marañón, Madrid, Spain

¹⁵ Servicio de Aparato Digestivo, Hospital Universitario Marqués de Valdecilla, Santander, Spain

¹⁶ Servicio de Aparato Digestivo, Hospital Universitario La Paz, Madrid, Spain

¹⁷ Gastroenterology and Multivisceral Transplant Unit, Department of Surgery, Oncology, and Gastroenterology, Padua University Hospital, Padua, Italy

¹⁸ Liver Research Center, Beijing Friendship Hospital, Capital Medial University, Beijing, China

¹⁹ Beijing Key Laboratory of Translational Medicine on Liver Cirrhosis, Beijing, China

²⁰ National Clinical Research Center for Digestive Diseases, Beijing, China

²¹ Department of Gastroenterology, Nara Medical University, Nara, Japan

²² Department of Gastroenterology and Hepatology, Antwerp University Hospital, Antwerp, Belgium

²³ Laboratory of Experimental Medicine and Paediatrics (LEMP), Faculty of Medicine and Health Sciences, University of Antwerp, Antwerp, Belgium

²⁴ InflaMed Centre of Excellence, University of Antwerp, Antwerp, Belgium

²⁵ Translational Sciences in Inflammation and Immunology, University of Antwerp,

Antwerp, Belgium

²⁶ Sheila Sherlock Liver Centre, Royal Free Hospital, London, United Kingdom

²⁷ Universidad Complutense de Madrid, Madrid, Spain

²⁸ Department of Visceral Surgery and Medicine, Inselspital, University of Bern, Bern,

Switzerland

Johngipter

FOOTNOTE PAGE					
Corresponding author:	Juan Carlos GARCÍA-PAGÁN, M.D. Ph.D.				
	Barcelona Hepatic Hemodynamic Lab, Liver Unit, Hospital				
	Clínic				
	Carrer de Villaroel, 170,				
	08036 Barcelona, Spain				
	Phone: +34 932275790				
	Fax: +34 932279856				
	E-Mail: jcgarcia@clinic.cat				
	Mattias MANDORFER, M.D. Ph.D.				
	Division of Gastroenterology and Hepatology,				
	Department of Internal Medicine III,				
	Medical University of Vienna,				
	Währinger Gürtel 18-20,				
	1090, Vienna, Austria				
	Phone: +43 1 40400 47440				
	Fax: +43 1 40400 47350				
	E-Mail: mattias.mandorfer@meduniwien.ac.at				
Key words:	HVPG; hepatic venous pressure gradient; CSPH; LSM;				
	liver stiffness measurement; transient elastography;				
	platelet count; NIT; chronic hepatitis C; sustained virologic				
response; SVR; aetiological cure;					

Title:

113 characters

	Journal Pre-proof
Abstract:	274
Total:	5684 including references
References:	41
Tables:	2
Figures:	6
Supplementary tables:	9
Supplementary figures:	6
Conflict of interest:	G.S. received travel support from Gilead.
	S.L. received grant support from Gilead and served as a
	speaker and/or consultant and/or advisory board member
	for AbbVie, Gilead, and MSD.
	E.L.M. received grant support from Novartis.
	A.B. has nothing to disclose.
	E.A. has nothing to disclose.
	E.L. has nothing to disclose.
	L.T. served as a speaker and/or consultant and/or advisory
	board member for AbbVie, Gilead, Bayern, Janssen, and
	W. L. Gore & Associates, and received travel support from
	AbbVie, MSD, and Gilead.
	P.S. served as a consultant for PharmalN.
	E.M. has nothing to disclose.
	L.E. has nothing to disclose.
	C.D. has nothing to disclose.
	L.IM. has nothing to disclose.
	A.P. has nothing to disclose.
	6

J.I.F. has nothing to disclose.

M.A. has nothing to disclose.

A.Z. has nothing to disclose.

A.C. has nothing to disclose.

H.H.-E. has nothing to disclose.

I.S.L.S. has nothing to disclose.

J.J. has nothing to disclose.

H.Y. served as a speaker and/or consultant and/or advisory board member for AbbVie, Gilead, Otuka, Asuka, and MSD. S.M.F. has nothing to disclose.

E.A.T. served as a speaker and/or consultant and/or advisory board member for Intercept, Gilead, Pfizer, NovoNordisk and Orphalan.

F.P.R. served as a speaker and/or consultant and/or advisory board member for AbbVie, Biotest, Gilead, and MSD, and received travel support from AbbVie, Biotest, and Gilead.

G.C. has nothing to disclose.

X.F. served as a speaker and/or consultant for AbbVie and Gilead.

R.B. served as a speaker and/or consultant and/or advisory board member for AbbVie, Gilead, and Janssen.

C.V. has nothing to disclose.

V.H.-G. served as a speaker and/or consultant and/or advisory board member for W. L. Gore & Associates.

T.R. served as a speaker and/or consultant and/or advisory board member for AbbVie, Bayer, Boehringer Ingelheim, Gilead, Intercept, MSD, Siemens, and W. L. Gore & Associates and received grants/research support from AbbVie, Boehringer Ingelheim, Gilead, MSD, Philips, and W. L. Gore & Associates as well as travel support from Boehringer Ingelheim and Gilead.

J.B. served as a lecturer or consultant for W. L. Gore & Associates, Actelion and Surrozen, and received grants from the 'Siftung für Leberkrankheiten Bern'.

J.C.G.-P. served as a speaker and/or consultant and/or advisory board member for Cook and W. L. Gore & Associates and received grants/research support from Conatus, Exalenz, Novartis, and Theravance.

M.M. served as a speaker and/or consultant and/or advisory board member for AbbVie, Bristol-Myers Squibb, Gilead, Collective Acumen, and W. L. Gore & Associates and received travel support from AbbVie, Bristol-Myers Squibb, and Gilead.

Financial support: This work was supported by a grant from the Medical Scientific Fund of the Major of the City of Vienna (No. 17035) as well as the Andrew K. Burroughs short-term training fellowship of the European Association for the Study of the Liver.

Author contributions: Study concept and design (G.S., S.L., J.C.G.-P., and M.M.), acquisition of data (all authors), analysis and interpretation of data (G.S., S.L., E.M., J.C.G.-P., and M.M.), drafting of the manuscript (G.S., S.L., J.C.G.-P., and M.M.), critical revision of the manuscript for important intellectual content (all authors).

Journal Pre-proof

ABSTRACT

Background & Aims: Non-invasive tests (NIT)s for clinically significant portal hypertension (CSPH; hepatic venous pressure gradient [HVPG] \geq 10mmHg) have predominantly been studied in patients with active HCV-infection. Investigations after HCV-cure are limited and yielded conflicting results. We conducted a pooled analysis to determine the diagnostic/prognostic utility of liver stiffness-measurement (LSM)/platelet count (PLT) in this setting.

Methods: 418 patients with pre-treatment HVPG≥6mmHg who achieved sustained virological response (SVR) and underwent post-treatment-HVPG-measurement were assessed, of which 324 (**HVPG/NIT-cohort**) also had paired data on pre-/post-treatment-LSM/-PLT.

The derived LSM/PLT-criteria were then validated against the direct endpoint decompensation in 755 compensated advanced chronic liver disease (cACLD) patients with SVR (cACLD-validation-cohort).

Results: HVPG/NIT-cohort: Among cACLD patients, the pre-/post-treatment prevalence of CSPH was 80%/54%. The correlation between LSM/HVPG increased from pre- to post-treatment (r=0.45 vs. 0.60), while that of PLT/HVPG remained unchanged. For given LSM/PLT-values, HVPG tended to be lower post- vs. pre-treatment, indicating the need for dedicated algorithms. Combining post-treatment-LSM/-PLT yielded a high diagnostic accuracy for post-treatment-CSPH in cACLD (AUC: 0.884 [95%CI: 0.843-0.926]). Post-treatment-LSM<12kPa & PLT>150G/L excluded CSPH (sensitivity: 99.2%), while LSM≥25kPa was highly specific for CSPH (93.6%).

cACLD-validation-cohort: The LSM<12kPa & PLT>150G/L-criterion was achieved in 42.5% of patients and their 3-year decompensation risk was 0%. In patients with post-

treatment-LSM≥25kPa (prevalence: 16.8%), 3-year decompensation risk was 9.6%, while it was 1.3% in those meeting none of the above criteria (prevalence: 40.7%). **Conclusions:** NITs can estimate the probability of CSPH after HCV-cure and predict clinical outcomes. cACLD patients with LSM<12kPa & PLT>150G/L (CSPH-excluded; no decompensation risk) may be discharged from portal hypertension surveillance (NITs and/or endoscopy), if no co-factors are present, while patients with LSM≥25kPa require surveillance/treatment (CSPH-ruled-in; increased decompensation risk).

LAY SUMMARY

Measurement of liver stiffness by a specific ultrasound device and platelet count (a simple blood test) are broadly used for the non-invasive diagnosis of increased blood pressure in the veins leading to the liver, which drives the development of complications in patients with advanced liver disease. The results of our pooled analysis refute previous concerns that these tests are less accurate after the cure of hepatitis C virus (HCV) infection. We have developed diagnostic criteria that facilitate the personalized management after HCV-cure and allow for a de-escalation of care in a high proportion of patients, thereby decreasing disease burden.

INTRODUCTION

Portal hypertension (PH) is the key driver of hepatic decompensation in patients with advanced chronic liver disease (ACLD) [1]. Accordingly, interventions that ameliorate portal hypertension have been shown to prevent hepatic decompensation in patients who are at risk, i.e., those with clinically significant portal hypertension (CSPH), which is defined by an HVPG ≥10mmHg. In addition to non-selective beta-blockers (NSBB) [1, 2], removal/suppression of the primary aetiological factor may lead to substantial reductions in HVPG, thereby decreasing the risk of hepatic decompensation. With the availability of interferon (IFN)-free regimens, sustained virological response (SVR; i.e., HCV-cure) is achieved in nearly all patients, despite the presence of pre-treatment ACLD and CSPH [3]. Previous studies in patients achieving SVR have reported an amelioration of PH across all pre-treatment HVPG strata [4-9]. In those with pretreatment CSPH, HVPG-decreases ≥10% were achieved in 60-63% [5-7]. However, only the absence/resolution of CSPH eliminates the risk of post-treatment hepatic decompensation, and thus, identifies patients who should be considered for deescalation of care to avoid unnecessary investigations and costs. The latter has profound economic implications, as the number of individuals who will achieve HCVcure world-wide is expected to exceed 1 million per year for the next decade, with a relevant proportion having compensated ACLD (cACLD) [10]. On the other end of the disease severity spectrum, those with post-treatment CSPH may remain at considerable risk. Since HVPG-measurement is invasive, resource-intensive, and requires considerable expertise [11, 12], CSPH risk stratification by non-invasive tests (NIT)s is key to individualize post-treatment management in patients with cACLD [13]. Platelet count (PLT) and liver stiffness measurement (LSM) by vibration-controlled transient elastography (VCTE) are the most extensively studied NITs for CSPH in cACLD patients [14, 15] and have been implemented in clinical practice

recommendations for the management of PH with Baveno VI [16]. However, their diagnostic ability for CSPH has predominantly been studied in patients with active HCV-infection, while investigations after HCV-cure are limited and yielded conflicting results [5-7, 17], which has led to considerable scepticism regarding their clinical use in this steadily increasing patient population [18].

Thus, we conducted a pooled analysis to investigate (i) the diagnostic performance of NITs for CSPH (primary objective) as well as (ii) the relationship between NITs and pre- and post-treatment HVPG and (iii) to validate the derived LSM/PLT criteria against the direct endpoint of hepatic decompensation (secondary objectives).

In addition, we **(iv)** described the evolution of PH after HCV-cure and **(v)** evaluate the diagnostic utility of NITs for varices and **(vi)** the relationship between PH and *de-novo* hepatocellular carcinoma (HCC) development.

METHODS

Patients

HVPG-cohort

After removing duplicates, 675 individual patients from 8 cohorts investigating HVPG in patients undergoing HCV-treatment (both IFN-containing and IFN-free) were evaluated for inclusion in this pooled analysis (Fig. 1) [4-9, 17, 19-22]. Information on exclusion criteria and patient selection is provided in Fig. 1. Authors of the 3 additional studies that have been published until 2020 were contacted, however, did not provide individual patient data [23-25]. Specifically, patients without paired HVPG-measurements before (baseline [BL]) and after (follow-up [FU]) HCV-treatment (n=166), without SVR (n=59), or without pre-treatment PH (i.e., HVPG <6mmHg) (n=32) were excluded. Finally, 418 patients were included to study the dynamics of HVPG after HCV-cure (HVPG-cohort). In patients undergoing repeated HVPG-assessments after HCV-cure, only the first post-treatment measurement was considered. All contributing centres had previously established protocols for measuring HVPG, which are in line with the technical description provided in the Baveno VII consensus [26].

HVPG/NIT-cohort

For the <u>evaluation of the diagnostic performance of NITs for PH</u>, 94 patients were additionally excluded due to missing information on either PLT or LSM (performed by vibration-controlled transient elastography [FibroScan, Echosens, Paris, France]) before and/or after HCV-treatment, resulting in 324 patients with paired data on HVPG/NITs pre- and post-treatment (**HVPG/NIT-cohort**). While we focused on the subgroup of patients with cACLD (i.e., absence/no history of hepatic decompensation defined by clinically evident ascites, portal-hypertensive bleeding, or overt hepatic

encephalopathy; n=241) and the overall cohort (i.e., including decompensated patients; n=324) in the main manuscript, subgroup analyses in patients with BL-CSPH (i.e., HVPG \geq 10mmHg; n=274), and clinical evidence of CSPH at BL (i.e., varices or presence/history of ascites/bleeding; n=184) can be found in the Supplementary materials.

The **HVPG/NIT-cohort** was further analysed in regard to the diagnostic utility of NITs for varices (n=201) and the relationship between PH and *de-novo* hepatocellular carcinoma (HCC; n=190) development.

Confirmation of the proposed criteria for excluding CSPH during long-term FU

To confirm the sensitivity/robustness of the proposed criteria for excluding CSPH based on NITs, they were additionally evaluated in a highly selected and particularly challenging series of 83 patients with pre-treatment CSPH in whom CSPH persisted 24 weeks after the end-of-treatment (EoT) and who underwent a third HVPG-measurement 96 weeks after EoT from Lens and Baiges et al (2020) [17].

Unselected cACLD-validation-cohort

Finally, two cohorts of cACLD patients (BL-LSM ≥10kPa, BL-HVPG ≥6mmHg, or advanced fibrosis/cirrhosis on liver histology [F3/4]) who have been followed for hepatic decompensation after achieving HCV-cure were included <u>to validate NITs</u> <u>against the direct endpoint</u>. The first cohort (n=368 from the Medical University of Vienna [MUV]) was partly overlapping (n=77) with the **HVPG-cohort**; the second multicentre cohort (n=387; centres Ordensklinikum Linz Barmherzige Schwestern, Padua University Hospital, and Klinikum Ottakring) was an entirely unrelated cohort recruited at centres which did not contribute to the **HVPG-cohort**. Of note, these patients have previously been reported in the context of hepatic decompensation [27] and/or HCC [28] risk stratification.

Statistics

Statistical analyses were performed using R 4.1.1 (R Core Team, R Foundation for Statistical Computing, Vienna, Austria). Continuous variables were reported as mean ±standard deviation or median (interguartile range), while categorical variables were reported as absolute and relative frequencies of patients with/without a certain characteristic. Repeated measures Pearson's correlation coefficients (to determine correlation coefficients within centres) and a summary coefficient were calculated using the 'rmcorr'-package [29]. Local regression (locally estimated scatterplot smoothing; LOESS) was applied as a non-parametric approach to visualize the relationship between two continuous variables (HVPG and LSM/PLT). The default span was α =0.75. Linear mixed effects regression models were fitted to study the relationship between NITs and HVPG, logistic mixed effects regression models to analyse the relationship between NITs and CSPH (HVPG ≥10mmHg) before and after HCV-cure using the 'Im4'-package [30]. For both approaches, NITs were treated as linear variables and modelled using natural B splines (i.e., restricted cubic splines) using the 'splines'-package to account for non-linear effects. Specifically, splines with 4 degrees of freedom were used. For linear regressions, marginal R² according to Nakagawa et al [31] were provided as a goodness-of-fit measure. For logistic regressions, the 'Akaike information criterion' (AIC) was reported. Relationships between CSPH and NITs were visualized using the 'effects'-package [32].

The area under the curve (AUC) and respective 95% confidence intervals (95%CI) of receiver operating characteristic (ROC) analyses were calculated based on predictions from fixed effects of respective mixed effects models adjusting for clustering across

centres as random term using the 'pROC'-package [33]. For prediction of CSPH after HCV-cure (FU-CSPH), PLT-values were capped if >150G/L to account for non-linearity above this cut-off, while LSM-values were log-transformed to account for non-normal-distribution [14]. These results from ordinary logistic regression analyses were compared to other methods popular in machine learning such as penalized regressions, naive Bayes and neural nets. Finally, nomograms and 3-D plots were developed using the 'rms'-package to provide clinically applicable prediction tools for FU-CSPH.

Fine & Gray competing risk regression analysis was performed for the outcomes HCC (**HVPG/NIT-cohort**) and hepatic decompensation (**cACLD-validation-cohort**) using the 'cmprsk'-package and cumulative incidence curves were plotted.

A two-sided p-value <0.05 was considered statistically significant for all analyses.

RESULTS

Evolution of PH after HCV-cure in the HVPG-cohort

Among 418 patients with paired HVPG-measurements, mean BL-HVPG was 14.2 ± 4.8 mmHg corresponding to 353 (84 %) patients with BL-CSPH and 153 (37%) with BL-HVPG \geq 16mmHg.

Median time between end-of-treatment (EoT) and post-treatment HVPG (FU-HVPG) was 28.4 (24-44) weeks (Supplementary Fig. 2).

HVPG decreased in 333 patients (80%), remained stable in 23 (5.5%) patients, and increased in 62 (14.8%), resulting in a mean FU-HVPG of 11.8±5.4mmHg. The median absolute difference between BL-HVPG and FU-HVPG was -2.5 (-4.3-[-0.50])mmHg, while the relative difference was -18.8 (-32.8-[-4.8])%.

Of 65 patients (16%) with BL-HVPG 6-9mmHg, 44 (68%) resolved PH (i.e., FU-HVPG <6mmHg) while 21 (32%) still had PH at FU (Fig. 2). Importantly, no patient progressed to FU-CSPH. Of 353 with BL-CSPH, 12 (3.4%) resolved PH, and 75 (21%) decreased to 6-9mmHg while 266 (75%) still had FU-CSPH. An HVPG-decrease \geq 10% was observed in 226/353 (64%) of patients with BL-CSPH. In the subgroup of patients with a BL-HVPG of 10-15mmHg (n=200, 48%), 80 (40%) resolved CSPH, and 9 (0.5%) progressed to HVPG \geq 16mmHg. Among patients with a BL-HVPG \geq 16mmHg (n=153, 37%), 7 (4.6%) resolved CSPH and 71 (46%) regressed to 10-15mmHg.

For analyses on the time-dependency of the relative decrease in HVPG after EoT, see the Supplementary materials/Supplementary Fig.s 1 and 2.

Relationship between HVPG and NITs before and after HCV-cure in the HVPG/NIT-cohort

The median time between EoT and assessment of FU-HVPG was 28.8 (IQR: 25-45) weeks. While the median time between laboratory assessment and FU-HVPG was 0

(IQR: 0-6) weeks, the median time period between FU-LSM and FU-HVPG was 0.6 (IQR: 0-14) weeks.

Importantly, comparisons of patient characteristics of the **HVPG/NIT-cohort** and the **HVPG-cohort** revealed no relevant differences (Supplementary Table 1). Characteristics of the individual subgroups of the **HVPG/NIT-cohort** are shown in Supplementary Table 2. Supplementary Fig. 3 shows scatterplots of HVPG and NITs before and after HCV-cure with regression lines for individual centres as well as the pooled regression lines/coefficients.

Of note, we observed an association of moderate strength between BL-PLT and BL-HVPG (summary r=-0.39) as well as BL-LSM and BL-HVPG (r=0.45) in the cACLD subgroup. Interestingly, the correlation of LSM/HVPG was stronger post-treatment (r=0.60), while it remained unchanged for PLT/HVPG. These observations were confirmed in the overall cohort.

Interestingly, the relationship between PLT/HVPG was nearly linear both at BL and FU, with lower predicted HVPG at FU than at BL given the same PLT count (Fig. 3). Thus, applying a PLT-based model/PLT cut-off derived in patients with active HCV-infection to patients who have achieved HCV-cure may result in an overestimation of the severity of PH. For LSM/HVPG, no linear relationship was observed at both time points, with an inflection point at 15-20kPa. Interestingly, after having achieved HCV-cure, lower levels of HVPG were observed for the same LSM until ~15kPa, i.e., in the range that is relevant for ruling-out/excluding CSPH. Even for values up to ~30kPa, LSM tended to overestimate the severity of PH after HCV-cure, although the 95%CIs were clearly overlapping.

Moreover, we investigated the explanatory ability, i.e., what proportion of the variance can be explained by PLT and LSM before and after treatment (Supplementary Table 3). LSM explained more variance in HVPG than PLT both before and after HCV-

treatment, and the application of non-linear modelling increased the goodness-of-fit in the cACLD subgroup and the overall cohort. Of note, combined models of PLT+LSM yielded the best explanatory ability, which was even numerically higher at FU, as compared to BL.

Estimating the probability of CSPH using NITs in the HVPG/NIT-cohort

After HCV-treatment, the prevalence of CSPH was 64% in the overall cohort and 54% in the cACLD subgroup. Again, the predictive ability was higher for LSM than for PLT (Supplementary Table 4; indicated by a lower AIC relative to each other). Of note, nonlinear modelling improved FU- but not BL-models (predicted probability of FU-CSPH based on FU-PLT and FU-LSM using natural B splines are illustrated in Supplementary Fig. 4).

ROC curves with respective AUCs are shown in Fig. 4. Of note, the AUC was similar/numerically higher at FU compared to BL. Although FU-LSM already yielded a high accuracy in the cACLD subgroup (AUC: 0.837 [95%CI: 0.786-0.887]), the AUCs showed numerical increases when combining FU-PLT and FU-LSM (AUC: 0.876 [95%CI: 0.831-0.920]). Comparable results were obtained in the overall cohort with an AUC of 0.895 (95%CI: 0.860-0.932) for FU-PLT+FU-LSM. Capping PLT-values at 150G/L and log-transforming LSM-values slightly increased diagnostic accuracy (cACLD subgroup AUC: 0.884 [95%CI: 0.843-0.926]). Of note, more complex models or methods popular in machine-learning did not increase the predictive ability (data not shown). To increase the clinical applicability of our model, nomograms and 3-D plots for estimating the probability of post-treatment CSPH based on post-treatment LSM/PLT in individual cACLD patients are provided (Fig. 5).

Moreover, similar graphs for the overall cohort are provided in the Supplementary Fig. 5. Further subgroup analyses can be found in Supplementary Table 5. Of note, the

diagnostic accuracies for CSPH (i.e., AUCs) were comparable, indicating the generalizability of our findings to these specific subgroups.

<u>Notably, the timing of FU-HVPG/NIT-assessment had no major impact on the</u> <u>diagnostic accuracy of NITs</u>. Detailed information is provided in the Supplementary materials.

Criteria to exclude and rule-in post-treatment CSPH in the HVPG/NIT-cohort

Potential criteria to rule-out FU-CSPH (different cut-offs for LSM and PLT as well as combinations such as the Baveno VI criteria and RECIST-HCV [34]) were compared in Table 2. Based on these data, post-treatment LSM <12kPa & PLT >150G/L was chosen for excluding post-treatment CSPH in cACLD patients (FU-CSPH prevalence in those meeting this criterion: 3.3%), as the sensitivity of not meeting this criterion was 99.2%. In contrast, post-treatment LSM ≥25kPa was highly specific (93.6%) for CSPH after HCV-cure, with a FU-CSPH prevalence of 87.7% in those meeting this criterion.

Confirmation of the proposed criteria for excluding CSPH during long-term FU

In brief, post-treatment LSM <12kPa & PLT >150G/L was capable of excluding FU-CSPH in a highly selected cohort of patients who were evaluated 96 weeks after EoT. See Supplementary materials for detailed information.

Validation against direct endpoints in the unselected cACLD-validation-cohort

The median durations from EoT to post-treatment evaluation were 11.6 (0.4-19.3) and 13.4 (12.0-26.6) weeks in the MUV and external cACLD-validation-cohorts, respectively, and patients were followed for clinical events for 55.4 (51.5-59.4) and 24.4 (21.0-27.5) months after EoT.

The LSM <12kPa & PLT >150G/L criterion was achieved in 43.2% and 41.9% in the MUV and external cACLD-validation-cohorts (Supplementary Table 7), respectively; the 3-year hepatic decompensation risk in patients meeting this criterium was 0% in both cohorts (Fig. 6). In patients with post-treatment LSM \geq 25kPa (prevalence: 16.8% and 16.8%), 3-year hepatic decompensation risk was 9.0%/11.0%, while it was 1.6%/0.8% in those meeting none of the above criteria (prevalence: 39.9% and 41.3%). When merging these two cohorts to a single **cACLD-validation-cohort** (n=755), 3-year hepatic decompensation risk was 0% in the 42.5% of patients who met the LSM <12kPa & PLT >150G/L criterion, 9.6% in the 16.8% with post-treatment LSM \geq 25kPa, and 1.3% in the 40.7% who met none of the above criteria (i.e., the diagnostic gray-zone in which CSPH can neither be excluded nor ruled-in).

Role of PH for de-novo HCC risk stratification in the HVPG-cohort

See Supplementary materials.

Use of NITs for ruling-out (large) varices in the HVPG-cohort

See Supplementary materials.

DISCUSSION

In this pooled analysis, we have synthesized the data of individual studies to provide robust information on the relationship between NITs and HVPG after HCV-cure. We have developed clinically useful tools for estimating the probability of CSPH and established that NITs are capable of excluding and ruling-in CSPH in the majority (i.e., 59.3%) of unselected patients with cACLD who have achieved SVR. The same criteria may be applied for non-invasive risk stratification. Finally, our pooled analysis may close a chapter ('evolution of PH after HCV-cure'), as it provides a comprehensive synthesis of the available data.

At a median of 28 weeks, we observed that HCV-cure was associated with a decrease in HVPG in ~80% of patients (on average: -18.8%) and none of the patients with pretreatment subclinical PH progressed to CSPH. We cannot discard that the changes observed at this time point may primarily reflect the amelioration of hepatic inflammation [20], as liver fibrosis regression may require long-term follow-up [35]. When evaluating the association between NITs and HVPG, we found a stronger correlation between LSM/HVPG after HCV-cure, as compared to patients with active HCV-infection, while no changes in the correlation between PLT/HVPG were

observed. The observation of an increasing strength of the association between LSM/HVPG but not PLT/HVPG is also reflected by an increase in the goodness-of-fit (as measured by R²) after HCV-cure for LSM but not PLT. The findings regarding LSM/HVPG may be explained by the treatment-induced decrease in PH severity, as the correlation between HVPG and LSM has previously been shown to be stronger in patients with values <10-12mmHg and weaker above this threshold [36-38]. This hypothesis is also supported by the lower R² for LSM observed in the subgroups of patients with pre-treatment CSPH (as assessed by HVPG/clinical findings) as these subgroups have a higher severity of PH.

Recent EASL guidelines on NITs urged for further studies/evidence regarding the use of NITs after SVR [18]. Importantly, our data refute the concern/paradigm, that LSM/PLT are less accurate for diagnosing CSPH in patients who have attained SVR. However, modelling the relationship of pre-/post-treatment LSM and PLT with pre-/post-treatment HVPG revealed that generalizing models that have been derived in patients with active HCV-infection would result in an overestimation of post-treatment HVPG by LSM/PLT. Accordingly, the estimated HVPG for a given LSM/PLT within the range that is relevant for clinical decision making was lower at FU, as compared to BL. These data clearly indicate that LSM/PLT-based models and algorithms specifically designed for patients who have achieved HCV-cure are required, highlighting the significance of our LSM/PLT-based criteria.

Excluding and ruling-in of CSPH has wide-ranging consequences, as patients without CSPH may be discharged from PH surveillance, while those with CSPH remain at considerable risk. Thus, we have analyzed this potential clinical application of NITs in more detail. Our study indicates that in cACLD patients, the diagnostic performance of LSM/PLT for CSPH was comparable or tended to be even better after HCV-cure, as compared to pre-treatment (AUC 0.753 vs. 0.800 for PLT, 0.831 vs. 0.837 for LSM and 0.871 vs 0.884 for the combination of both). This underscores their utility after HCV-cure and possibly also after the suppression/removal of other primary aetiological factors – a clinical scenario that is becoming increasingly common due to considerable progress in the field of aetiological therapies. Focusing on potential clinical utility, the nomograms and 3-D plots derived from our study allow to estimate the probability of CSPH in a given patient. To simplify clinical decision making, we developed criteria for excluding and ruling-in CSPH. Based on our data, Post-treatment LSM <12kPa and PLT >150G/L showed a sensitivity of 99.2% for CSPH, and thus, would be the best criterion to exclude CSPH. The sensitivity/robustness of this criterium was further

confirmed in a highly selected cohort of patients who were evaluated 96 weeks after EoT and in whom previous research indicated that LSM <13.6kPa is incapable of ruling-out CSPH, thereby fuelling the debate on the utility of NITs for CSPH after HCVcure [17]. However, lowering the LSM cut-off, but also considering PLT substantially increased the sensitivity. The significance of PLT in this context may be explained by persistence of histological lesions after HCV-cure [39, 40], which may lead to low LSM despite CSPH. However, persistent thrombocytopenia may still reveal CSPH in these patients. Of note, Baveno VI criteria also vielded a high sensitivity for post-treatment CSPH (94.7%), with a CSPH prevalence of only 14% in patients meeting these criteria and no large varices being missed. While RECIST-HCV criteria [34] (which do not require LSM) identified a substantially higher proportion of patients being at low risk and did not miss any patient with large varices, the CSPH prevalence in the low-risk group was comparatively high (28%) and we also observed hepatic decompensation events in the cACLD-validation-cohort. Accordingly, these criteria were suboptimal for identifying patients who may be safely discharged from PH surveillance. LSM ≥25kPa was highly specific for CSPH (93.1%), with a post-treatment CSPH prevalence of 87.7% - i.e., CSPH can be ruled-in in these patients, arguing for maintaining carvedilol/NSBB therapy. Although the probability of post-treatment CSPH can be estimated based on the provided nomogram, HVPG-measurement [11, 12] is the only method to ascertain the absence/presence of CSPH in patients within the gray-zone. Nevertheless, the risk of decompensation in the gray-zone was very low (1.3% at 3 years) indicating that future studies comprising even higher numbers of patient years may help to broaden the low-risk group by identifying additional patients in whom the risk of hepatic decompensation is negligible.

Since the LSM <12kPa and PLT >150G/L criterion will be applied to identify patients without post-treatment CSPH (i.e., a surrogate endpoint) who are candidates for being

discharged from PH surveillance, this decision rule needed to be thoroughly validated against direct outcomes (i.e., hepatic decompensation). Therefore, we included two large cohorts of cACLD patients (cACLD-validation-cohort); one of them was an entirely unrelated cohort recruited at academic but also non-academic centres which did not contribute to the other cohorts. This approach confirmed that no hepatic decompensation occurred in patients meeting these criteria when considering HCC development and death as competing events. Accordingly, there is no room for preventive strategies (no risk – no risk reduction achievable) in terms of hepatic decompensation in cACLD patients meeting these criteria, which strongly argues for their discharge from PH surveillance (NITs and/or endoscopy), if improvements in NITs are consistent and no co-factors are present. This finding has important practical implications for descalating care – an aspect that has largely been neglected by previous research, but seems to be crucial for regaining quality of life and decreasing resource utilization. Importantly, risk stratification approaches for post-treatment HCC differ [28], and thus, discontinuation of NITs and/or endoscopy does not include discontinuation of HCC screening. While we observed that de-novo HCC development was more common in patients with CSPH after HCV-cure, HVPG did not improve risk stratification on top of previously established non-invasive algorithms [28].

We must acknowledge several limitations of our study. First, the **HVPG/NIT-cohort** is not fully representative for the spectrum of (c)ACLD patients achieving HCV-cure, since due to the intrinsic characteristics of the included studies, the pre-treatment severity of PH was high. Accordingly, the proportions of patients meeting non-invasive criteria and information regarding the sensitivity/specificity as well as the proportion of patients having CSPH within different strata have to be interpreted with caution, as they directly depend on the prevalence of CSPH. However, the CSPH prevalence after HCV-cure was comparable to the ANTICIPATE study [14]. Moreover, unbiased

estimates regarding the proportion of patients meeting the non-invasive criteria are provided in the two separate cACLD cohorts and the event rates in the LSM <12kPa & PLT >150G/L and LSM ≥25kPa were in line with expectations – no risk in those in whom CSPH can be excluded [7, 17] and increased risk of hepatic decompensation in those in whom CSPH was ruled-in [7, 17]. However, event rates in those with ≥25kPa seemed to be lower than those of CSPH patients included in previous studies [2, 41]. Due to the removal of the primary aetiological factor, LSM (and HVPG) are likely to have decreased over time in our patients, thereby further mitigating the risk of hepatic decompensation. Accordingly, reassessments of LSM at later time points may have re-classified patients and provided more accurate prognostic information, which is a second limitation of our study. Third, we cannot rule-out entirely that the association between NITs and HVPG differs, if re-evaluated at late time points. However, stratifying by time from EoT to HVPG/NITs, there was no clear evidence for a decreasing diagnostic ability of NITs for CSPH throughout the time from EoT to NIT-strata.

In conclusion, the results of this pooled analysis refute the previous concern that NITs are less capable of staging PH after HCV-cure. Indeed, NITs can estimate the probability/exclude/rule-in CSPH after HCV-cure and predict clinical outcomes. Based on these findings, Baveno VII [26] recommends that patients with LSM <12kPa & PLT >150G/L (CSPH-excluded; no decompensation risk) may be discharged from portal hypertension surveillance (NITs and/or endoscopy), if no co-factors are present, while continuation of carvedilol may be considered in those with LSM ≥25kPa (CSPH-ruled-in; increased decompensation risk).

ABBREVIATIONS:	95%CI 95% confidence interval				
	ACLD advanced chronic liver disease				
	aSHR adjusted subdistribution hazard ratio				
	AUC area under the curve				
	BL baseline				
	cACLD compensated ACLD				
	CSPH clinically significant portal hypertension				
	EOT end-of-treatment				
	FU follow-up				
	HCC hepatocellular carcinoma				
	HVPG hepatic venous pressure gradient				
	IQR interquartile range				
	LSM liver stiffness measurement				
	MUV Medical University of Vienna				
	NIT non-invasive test				
	NSBB non-selective beta-blocker				
	PLT platelet count				
	ROC receiver operating characteristic				
	SHR subdistribution hazard ratio				
	SVR sustained virologic response				

ACKNOWLEDGEMENTS

We would like to acknowledge the contributions of Stefanie HAMETNER-SCHREIL, and Rainer SCHÖFL (Ordensklinikum Linz Barmherzige Schwestern) as well as Michael SCHWARZ, Caroline SCHWARZ, and Michael GSCHWANTLER (Klinikum Ottakring) to the external **cACLD-validation-cohort**.

Journal Prevention

REFERENCES

Author names in bold designate shared co-first authorship

[1] Mandorfer M, Simbrunner B. Prevention of First Decompensation in Advanced Chronic Liver Disease. Clin Liver Dis 2021;25:291-310.

[2] Villanueva C, Albillos A, Genesca J, Garcia-Pagan JC, Calleja JL, Aracil C, et al. beta blockers to prevent decompensation of cirrhosis in patients with clinically significant portal hypertension (PREDESCI): a randomised, double-blind, placebocontrolled, multicentre trial. Lancet (London, England) 2019;393:1597-1608.

[3] Mandorfer M, Kozbial K, Freissmuth C, Schwabl P, Stattermayer AF, Reiberger T, et al. Interferon-free regimens for chronic hepatitis C overcome the effects of portal hypertension on virological responses. Alimentary pharmacology & therapeutics 2015;42:707-718.

[4] Lens S, Rincón D, García-Retortillo M, Albillos A, Calleja JL, Bañares R, et al. Association Between Severe Portal Hypertension and Risk of Liver Decompensation in Patients With Hepatitis C, Regardless of Response to Antiviral Therapy. Clinical gastroenterology and hepatology : the official clinical practice journal of the American Gastroenterological Association 2015;13:1846-1853.e1841.

[5] **Mandorfer M, Kozbial K**, Schwabl P, Freissmuth C, Schwarzer R, Stern R, et al. Sustained virologic response to interferon-free therapies ameliorates HCV-induced portal hypertension. Journal of hepatology 2016;65:692-699.

[6] Lens S, Alvarado-Tapias E, Marino Z, Londono MC, E LL, Martinez J, et al. Effects of All-Oral Anti-Viral Therapy on HVPG and Systemic Hemodynamics in Patients With Hepatitis C Virus-Associated Cirrhosis. Gastroenterology 2017;153:1273-1283.e1271.

[7] Mandorfer M, Kozbial K, Schwabl P, Chromy D, Semmler G, Stättermayer AF, et al. Changes in Hepatic Venous Pressure Gradient Predict Hepatic Decompensation

in Patients Who Achieved Sustained Virologic Response to Interferon-Free Therapy. Hepatology (Baltimore, Md) 2020;71:1023-1036.

[8] Mauro E, Crespo G, Montironi C, Londoño MC, Hernández-Gea V, Ruiz P, et al. Portal pressure and liver stiffness measurements in the prediction of fibrosis regression after sustained virological response in recurrent hepatitis C. Hepatology (Baltimore, Md) 2018;67:1683-1694.

[9] Díez C, Berenguer J, Ibañez-Samaniego L, Llop E, Pérez-Latorre L, Catalina MV, et al. Persistence of Clinically Significant Portal Hypertension After Eradication of Hepatitis C Virus in Patients With Advanced Cirrhosis. Clinical infectious diseases : an official publication of the Infectious Diseases Society of America 2020;71:2726-2729.

[10] Heffernan A, Cooke GS, Nayagam S, Thursz M, Hallett TB. Scaling up prevention and treatment towards the elimination of hepatitis C: a global mathematical model. Lancet (London, England) 2019;393:1319-1329.

[11] Bosch J, Abraldes JG, Berzigotti A, García-Pagan JC. The clinical use of HVPG measurements in chronic liver disease. Nature Reviews Gastroenterology & Hepatology 2009;6:573-582.

[12] Reiberger T, Schwabl P, Trauner M, Peck-Radosavljevic M, Mandorfer M.Measurement of the Hepatic Venous Pressure Gradient and Transjugular Liver Biopsy.JoVE 2020:e58819.

[13] Semmler G, Binter T, Kozbial K, Schwabl P, Hametner-Schreil S, Zanetto A, et al. Non-invasive risk stratification after HCV-eradication in patients with advanced chronic liver disease. Hepatology 2020.

[14] Abraldes JG, Bureau C, Stefanescu H, Augustin S, Ney M, Blasco H, et al. Noninvasive tools and risk of clinically significant portal hypertension and varices in compensated cirrhosis: The "Anticipate" study. Hepatology (Baltimore, Md) 2016;64:2173-2184.

[15] Mandorfer M, Hernández-Gea V, García-Pagán JC, Reiberger T. Noninvasive Diagnostics for Portal Hypertension: A Comprehensive Review. Seminars in liver disease 2020;40:240-255.

[16] de Franchis R, Baveno VIF. Expanding consensus in portal hypertension: Report of the Baveno VI Consensus Workshop: Stratifying risk and individualizing care for portal hypertension. Journal of hepatology 2015;63:743-752.

[17] **Lens S, Baiges A**, Alvarado E, Llop E, Martinez J, Fortea JI, et al. Clinical outcome and hemodynamic changes following HCV eradication with oral antiviral therapy in patients with clinically significant portal hypertension. Journal of hepatology 2020.

[18] Berzigotti A, Boursier J, Castera L, Cazzagon N, Friedrich-Rust M, Petta S, et al. Easl Clinical Practice Guidelines (Cpgs) On Non-Invasive Tests For Evaluation Of Liver Disease Severity And Prognosis- 2020 Update. Journal of hepatology 2021.

[19] Reiberger T, Payer BA, Ferlitsch A, Sieghart W, Breitenecker F, Aichelburg MC, et al. A prospective evaluation of pulmonary, systemic and hepatic haemodynamics in HIV-HCV-coinfected patients before and after antiviral therapy with pegylated interferon and ribavirin. Antiviral therapy 2012;17:1327-1334.

[20] **Schwabl P, Mandorfer M**, Steiner S, Scheiner B, Chromy D, Herac M, et al. Interferon-free regimens improve portal hypertension and histological necroinflammation in HIV/HCV patients with advanced liver disease. Alimentary pharmacology & therapeutics 2017;45:139-149.

[21] Puente Á, Cabezas J, López Arias MJ, Fortea JI, Arias MT, Estébanez Á, et al. Influence of sustained viral response on the regression of fibrosis and portal hypertension in cirrhotic HCV patients treated with antiviral triple therapy. Revista espanola de enfermedades digestivas : organo oficial de la Sociedad Espanola de Patologia Digestiva 2017;109:17-25.

[22] Abadía M, Montes ML, Ponce D, Froilán C, Romero M, Poza J, et al. Management of betablocked patients after sustained virological response in hepatitis C cirrhosis. World journal of gastroenterology 2019;25:2665-2674.

[23] Roberts S, Gordon A, McLean C, Pedersen J, Bowden S, Thomson K, et al.
 Effect of Sustained Viral Response on Hepatic Venous Pressure Gradient in Hepatitis
 C–Related Cirrhosis. Clinical Gastroenterology and Hepatology 2007;5:932-937.

[24] Rincon D, Ripoll C, Lo Iacono O, Salcedo M, Catalina MV, Alvarez E, et al. Antiviral therapy decreases hepatic venous pressure gradient in patients with chronic hepatitis C and advanced fibrosis. The American journal of gastroenterology 2006;101:2269-2274.

[25] Afdhal N, Everson GT, Calleja JL, McCaughan GW, Bosch J, Brainard DM, et al. Effect of viral suppression on hepatic venous pressure gradient in hepatitis C with cirrhosis and portal hypertension. Journal of viral hepatitis 2017;24:823-831.

[26] de Franchis R, Bosch J, Garcia-Tsao G, Reiberger T, Ripoll C, Abraldes JG, et al. BAVENO VII - RENEWING CONSENSUS IN PORTAL HYPERTENSION: Report of the Baveno VII Consensus Workshop: personalized care in portal hypertension. Journal of hepatology.

[27] Semmler G, Binter T, Kozbial K, Schwabl P, Hametner-Schreil S, Zanetto A, et al. Noninvasive Risk Stratification After HCV Eradication in Patients With Advanced Chronic Liver Disease. Hepatology (Baltimore, Md) 2021;73:1275-1289.

[28] Semmler G, Meyer EL, Kozbial K, Schwabl P, Hametner-Schreil S, Zanetto A, et al. HCC risk stratification after cure of hepatitis C in patients with compensated advanced chronic liver disease. Journal of hepatology 2022;76:812-821.

[29] Bakdash JZ, Marusich LR. Repeated Measures Correlation. Frontiers in Psychology 2017;8.

[30] Bates D, Mächler M, Bolker B, Walker S. Fitting Linear Mixed-Effects Models Using Ime4. Journal of Statistical Software 2015;67:1 - 48.

[31] Nakagawa S, Johnson PCD, Schielzeth H. The coefficient of determination R(2) and intra-class correlation coefficient from generalized linear mixed-effects models revisited and expanded. Journal of the Royal Society, Interface 2017;14.

[32] Fox J, Weisberg S. Visualizing Fit and Lack of Fit in Complex Regression Models with Predictor Effect Plots and Partial Residuals. Journal of Statistical Software 2018;87:1 - 27.

[33] Robin X, Turck N, Hainard A, Tiberti N, Lisacek F, Sanchez J-C, et al. pROC: an open-source package for R and S+ to analyze and compare ROC curves. BMC Bioinformatics 2011;12:77.

[34] Calvaruso V, Cacciola I, Licata A, Madonia S, Benigno R, Petta S, et al. Is Transient Elastography Needed for Noninvasive Assessment of High-Risk Varices? The REAL Experience. Official journal of the American College of Gastroenterology | ACG 2019;114:1275-1282.

[35] Marcellin P, Gane E, Buti M, Afdhal N, Sievert W, Jacobson IM, et al. Regression of cirrhosis during treatment with tenofovir disoproxil fumarate for chronic hepatitis B: a 5-year open-label follow-up study. Lancet (London, England) 2013;381:468-475.

[36] Vizzutti F, Arena U, Romanelli RG, Rega L, Foschi M, Colagrande S, et al. Liver stiffness measurement predicts severe portal hypertension in patients with HCV-related cirrhosis. Hepatology (Baltimore, Md) 2007;45:1290-1297.

[37] Colecchia A, Montrone L, Scaioli E, Bacchi–Reggiani ML, Colli A, Casazza G, et al. Measurement of Spleen Stiffness to Evaluate Portal Hypertension and the Presence of Esophageal Varices in Patients With HCV-Related Cirrhosis. Gastroenterology 2012;143:646-654.

[38] Reiberger T, Ferlitsch A, Payer BA, Pinter M, Homoncik M, Peck-RadosavljevicM. Non-selective beta-blockers improve the correlation of liver stiffness and portal pressure in advanced cirrhosis. Journal of gastroenterology 2012;47:561-568.

[39] D'Ambrosio R, Aghemo A, Fraquelli M, Rumi MG, Donato MF, Paradis V, et al. The diagnostic accuracy of Fibroscan for cirrhosis is influenced by liver morphometry in HCV patients with a sustained virological response. Journal of hepatology 2013;59:251-256.

[40] Pan JJ, Bao F, Du E, Skillin C, Frenette CT, Waalen J, et al. Morphometry Confirms Fibrosis Regression From Sustained Virologic Response to Direct-Acting Antivirals for Hepatitis C. Hepatology communications 2018;2:1320-1330.

[41] Ripoll C, Groszmann R, Garcia-Tsao G, Grace N, Burroughs A, Planas R, et al. Hepatic venous pressure gradient predicts clinical decompensation in patients with compensated cirrhosis. Gastroenterology 2007;133:481-488.

FIGURE LEGENDS

platelet count

Fig. 1. Patient flow-chart. Abbreviations: BL baseline; cACLD compensated advanced chronic liver disease; CSPH clinically significant portal hypertension; HVPG hepatic venous pressure gradient; NIT non-invasive test

Fig. 2. The dynamics of HVPG from pre- (BL) to post-treatment (FU). Abbreviations: BL baseline; FU follow-up; HVPG hepatic venous pressure gradient

Fig. 3. Relationship between NITs (i.e., PLT and LSM) and HVPG before and after HCV-treatment in the HVPG/NIT-cohort modelled with local regression (LOESS). (A) cACLD subgroup (n=241), and (B) overall cohort (n=324). Red and blue lines indicate the relationship before and after HCV-cure, respectively. For this analysis, no adjustment for clustering was performed/possible. Abbreviations: BL baseline; cACLD compensated advanced chronic liver disease; FU follow-up; HVPG hepatic venous pressure gradient; LSM liver stiffness measurement; NIT non-invasive test; PLT

Fig. 4. ROC-curves of NITs to detect CSPH before and after HCV-cure. (A-C) cACLD subgroup (n=241), and (D-F) the overall HVPG/NIT-cohort (n=324). Abbreviations: AUC area under the curve; BL baseline; cACLD compensated advanced chronic liver disease; CSPH clinically significant portal hypertension; FU follow-up; HVPG hepatic venous pressure gradient; NIT non-invasive test; ROC receiver operator characteristics; 95%CI 95% confidence interval

Fig. 5. Nomogram and 3-D plot for the prediction of FU-CSPH based on FU-PLT and FU-LSM in the cACLD subgroup (n=241). FU-PLT were truncated at 150 G/L and FU-LSM log-transformed. Amplified graphs can be found in the supplementary materials. Abbreviations: cACLD compensated advanced chronic liver disease; CSPH clinically significant portal hypertension; FU follow-up; LSM liver stiffness measurement; PLT platelet count

Fig. 6. Cumulative incidence of hepatic decompensation in cACLD patients stratified according to non-invasive criteria for excluding (FU-LSM <12kPa & FU-PLT >150G/L) or ruling-in CSPH (FU-LSM ≥25kPa) after HCV-cure. (A) Medical University of Vienna (MUV-cohort, n=368), (B) independent tertiary centers (external cohort, n=387), and (C) a combined cACLD-validation-cohort (n=755). HCC and death were considered competing risks (dashed lines). Abbreviations: cACLD compensated advanced chronic liver disease; CSPH clinically significant portal hypertension; FU follow-up; HCC hepatocellular carcinoma; HVPG hepatic venous pressure gradient; LSM liver stiffness measurement; MUV Medical university of Vienna; NA not applicable; PLT platelet count; SHR subdistribution hazard ratio

TABLES

Table 1. AUC and respective performance metrics of NITs for CSPH before and after HCV-cure in the HVPG/NIT-cohort: cACLD subgroup (n=241; A), and the overall cohort (n=324; B). Predictions were obtained from fixed effects of the respective linear mixed effects models adjusting for the centre as random effect.

Variables	Outcome	AUC	95%CI
A – cACLD subgroup			
BL-PLT	BL-CSPH	0.753	0.677-0.828
BL-LSM	BL-CSPH	0.831	0.769-0.894
BL-PLT + BL-LSM	BL-CSPH	0.859	0.807-0.912
BL-PLT + BL-LSM – transformed ¹	BL-CSPH	0.871	0.819-0.923
FU-PLT	FU-CSPH	0.800	0.745-0.855
FU-LSM	FU-CSPH	0.837	0.786-0.887
FU-PLT + FU-LSM	FU-CSPH	0.876	0.831-0.920
FU-PLT + FU-LSM - transformed ¹	FU-CSPH	0.884	0.843-0.926
B – overall cohort			
BL-PLT	BL-CSPH	0.778	0.709-0.847
BL-LSM	BL-CSPH	0.854	0.800-0.908
BL-PLT + BL-LSM	BL-CSPH	0.883	0.840-0.926
BL-PLT + BL-LSM - transformed ¹	BL-CSPH	0.894	0.852-0.937
FU-PLT	FU-CSPH	0.823	0.777-0.869
FU-LSM	FU-CSPH	0.857	0.815-0.900
FU-PLT + FU-LSM	FU-CSPH	0.895	0.860-0.932
FU-PLT + FU-LSM - transformed ¹	FU-CSPH	0.902	0.868-0.936

¹PLT were capped at 150G/L if >150G/L and LSM were log-transformed;

Abbreviations:

AUC area under the curve

BL baseline

cACLD compensated advanced chronic liver disease

CSPH clinically significant portal hypertension

FU follow-up

HVPG hepatic venous pressure gradient

LSM liver stiffness measurement

NIT non-invasive test

PLT platelet count

Table 2. Prevalence and sensitivity/specificity of selected NIT-criteria for diagnosis of CSPH after HCV-cure in the HVPG/NIT-cohort: cACLD subgroup (n=241; A) and the overall cohort (n=324; B). Although the primary intention behind these criteria is to rule-out/exclude FU-CSPH, diagnostic indices are reported for FU-CSPH.

Variables	Patients	Prevalence	Prevalence	Sensitivity	Specificity
	meeting the	of FU-CSPH	of FU-CSPH	for FU-	for FU-
	criterion	when	when <u>not</u>	CSPH ³	CSPH ³
		meeting the	meeting the		
		criterion ¹	criterion ²		
A – cACLD subgroup (n=241)					
FU-LSM <10kPa	51 (21%)	7/51 (14%)	124/190	124/131	44/110
			(65%)	(95%)	(40%)
FU-LSM <12kPa	73 (30%)	13/73 (18%)	118/168	118/131	60/110
			(70%)	(90%)	(55%)
FU-LSM <15kPa	109 (45%)	26/109	105/132	105/131	83/110
		(24%)	(80%)	(80%)	(76%)
FU-LSM <20kPa	159 (66%)	62/159	69/82 (84%)	69/131	97/110
		(39%)		(53%)	(88%)
FU-LSM <25kPa	184 (76%)	81/184	50/57 (88%)	50/131	103/110
		(44%)		(38%)	(94%)
FU-PLT >150G/L	58 (24%)	12/58 (21%)	119/183	119/131	46/110
			(65%)	(91%)	(42%)
FU-LSM <12kPa and FU-PLT	30 (12%)	1/30 (3.3%)	130/211	130/131	29/110
>150G/L			(62%)	(99%)	(26%)
FU-LSM <15kPa and FU-PLT	42 (17%)	4/42 (9.5%)	127/199	127/131	38/110
>150G/L			(64%)	(97%)	(35%)
FU-LSM <20kPa and FU-PLT	50 (21%)	7/50 (14%)	124/191	124/131	43/110
>150G/L (Baveno VI criteria)			(65%)	(95%)	(39%)
FU-albumin >36g/L and FU-PLT	91 (39%)	25/91 (28%)	102/143	102/127	66/107
>120G/L (RECIST-HCV			(71%)	(80%)	(62%)
criteria) ⁴					
B – overall cohort (n=324)					
FU-LSM <10kPa	54 (17%)	9/54 (17%)	199/270	199/208	45/116
			(74%)	(96%)	(39%)
FU-LSM <12kPa	79 (24%)	16/79 (20%)	192/245	192/208	63/116
			(78%)	(92%)	(54%)
FU-LSM <15kPa	121 (37%)	34/121	174/203	174/208	87/116
		(28%)	(86%)	(84)%	(75%)
FU-LSM <20kPa	190 (59%)	88/190	120/134	120/208	102/116
		(46%)	(90%)	(58%)	(88%)
FU-LSM <25kPa	222 (69%)	114/222	94/102	94/208	108/116
		(51%)	(92%)	(45%)	(93%)
FU-PLT >150G/L	62 (19%)	15/62 (24%)	193/262	193/208	47/116
			(74%)	(93%)	(41%)
FU-LSM <12kPa and FU-PLT	31 (9.6%)	2/31 (6.5%)	206/293	206/208	29/116
>150G/L			(70%)	(99%)	(25%)
FU-LSM <15kPa and FU-PLT	43 (13%)	5/43 (12%)	203/281	203/208	38/116
>150G/L			(72%)	(98%)	(33%)
FU-LSM <20kPa and FU-PLT	53 (16%)	9/53 (17%)	199/271	199/208	44/116
>150G/L (Baveno VI criteria)			(73%)	(96%)	(38%)

Journal Pre-proof							
FU-albumin>36g/L and FU- PLT>120G/L (RECIST-HCV criteria) ⁴	102 (32%)	34/102 (33%)	169/214 (79%)	169/203 (83%)	68/113 (60%)		

¹ Corresponding to the false-negative rate or the reciprocal of the negative predictive value when not meeting the criterion; ² Corresponding to the positive predictive value for FU-CSPH when not meeting the respective criterion; ³ Calculated for patients not meeting the respective criterion; ⁴ Available in 234 cACLD patients and 316 in the overall cohort

Abbreviations:

BL baseline

cACLD compensated advanced chronic liver disease

FU follow-up

LSM liver stiffness measurement

PLT platelet count

ournalPro

TABLES

Table 1. AUC and respective performance metrics of NITs for CSPH before and after HCV-cure in the HVPG/NIT-cohort: cACLD subgroup (n=241; A), and the overall cohort (n=324; B). Predictions were obtained from fixed effects of the respective linear mixed effects models adjusting for the centre as random effect.

Variables	Outcome	AUC	95%CI
A – cACLD subgroup			
BL-PLT	BL-CSPH	<mark>0.753</mark>	0.677-0.828
BL-LSM	BL-CSPH	<mark>0.831</mark>	<mark>0.769-0.894</mark>
BL-PLT + BL-LSM	BL-CSPH	<mark>0.859</mark>	0.807-0.912
BL-PLT + BL-LSM – transformed ¹	BL-CSPH	<mark>0.871</mark>	0.819-0.923
FU-PLT	FU-CSPH	<mark>0.800</mark>	0.745-0.855
FU-LSM	FU-CSPH	<mark>0.837</mark>	0.786-0.887
FU-PLT + FU-LSM	FU-CSPH	<mark>0.876</mark>	<mark>0.831-0.920</mark>
FU-PLT + FU-LSM - transformed ¹	FU-CSPH	<mark>0.884</mark>	<mark>0.843-0.926</mark>
B – overall cohort		0	
BL-PLT	BL-CSPH	<mark>0.778</mark>	<mark>0.709-0.847</mark>
BL-LSM	BL-CSPH	<mark>0.854</mark>	<mark>0.800-0.908</mark>
BL-PLT + BL-LSM	BL-CSPH	0.883	<mark>0.840-0.926</mark>
BL-PLT + BL-LSM - transformed ¹	BL-CSPH	<mark>0.894</mark>	<mark>0.852-0.937</mark>
FU-PLT	FU-CSPH	<mark>0.823</mark>	<mark>0.777-0.869</mark>
FU-LSM	FU-CSPH	<mark>0.857</mark>	<mark>0.815-0.900</mark>
FU-PLT + FU-LSM	FU-CSPH	<mark>0.895</mark>	<mark>0.860-0.932</mark>
FU-PLT + FU-LSM - transformed ¹	FU-CSPH	<mark>0.902</mark>	<mark>0.868-0.936</mark>

¹PLT were capped at 150G/L if >150G/L and LSM were log-transformed;

Abbreviations:

AUC area under the curve

BL baseline

cACLD compensated advanced chronic liver disease

CSPH clinically significant portal hypertension

FU follow-up

HVPG hepatic venous pressure gradient

LSM liver stiffness measurement

NIT non-invasive test

PLT platelet count

Table 2. Prevalence and sensitivity/specificity of selected NIT-criteria for diagnosis of CSPH after HCV-cure in the HVPG/NIT-cohort: cACLD subgroup (n=241; A) and the overall cohort (n=324; B). Although the primary intention behind these criteria is to rule-out/exclude FU-CSPH, diagnostic indices are reported for FU-CSPH.

Variables	Patients	Prevalence	Prevalence	Sensitivity	Specificity
	meeting the	of FU-CSPH	of FU-CSPH	for FU-	for FU-
	criterion	when	when <u>not</u>	CSPH ³	CSPH ³
		meeting the	meeting the		
		criterion.	criterion-	_	
A = CACLD subgroup (n=241)		7/64 (440/)	124/100	104/101	44/440
FU-LSWI < TUKPa	<mark>01 (21%)</mark>	<mark>7/31 (14%)</mark>	124/190 (65%)	$\frac{124}{131}$	44/110 (40%)
FILLI SM ~12kPa	73 (30%)	13/73 (18%)	(05 %) 118/168	(3370) 118/131	(4070) 60/110
	10 (0070)	13/13 (1070)	(70%)	(90%)	(55%)
FU-LSM <15kPa	109 (45%)	26/109	105/132	105/131	83/110
		(24%)	(80%)	(80%)	(76%)
FU-LSM <20kPa	159 (66%)	62/159	69/82 (84%)	69/131	97/110
	, , , , , , , , , , , , , , , , , , ,	(39%)		(53%)	(88%)
FU-LSM <25kPa	184 (76%)	81/184	50/57 (88%)	50/131	103/110
		(44%)		(38%)	(94%)
FU-PLT >150G/L	58 (24%)	12/58 (21%)	119/183	119/131	46/110
			(65%)	(91%)	(42%)
FU-LSM <12kPa and FU-PLT	30 (12%)	1/30 (3.3%)	130/211	130/131	29/110
>150G/L			(62%)	(99%)	(26%)
FU-LSM <15kPa and FU-PLT	42 (17%)	4/42 (9.5%)	127/199	127/131	38/110
>150G/L			(64%)	(97%)	(35%)
FU-LSM <20kPa and FU-PLT	50 (21%)	7/50 (14%)	124/191	124/131	43/110
>150G/L (Baveno VI criteria)	04 (000())	05/04 (00%)	(65%)	(95%)	(39%)
	<mark>91 (39%)</mark>	<mark>25/91 (28%)</mark>	102/143 (719/)	$\frac{102}{127}$	66/107 (629()
sritoria)4			(71%)	(80%)	(02%)
B = overall cohort (n=324)					
$\frac{1}{1} = \frac{1}{1} = \frac{1}$	<u>51 (17%)</u>	0/51(17%)	100/270	100/208	15/116
		3/34 (17/0)	(74%)	(96%)	(39%)
FU-I SM <12kPa	79 (24%)	16/79 (20%)	192/245	192/208	63/116
	10 (2170)		(78%)	(92%)	(54%)
FU-LSM <15kPa	121 (37%)	34/121	174/203	174/208	87/116
		(28%)	(86%)	(84)%	(75%)
FU-LSM <20kPa	190 (59%)	88/190	120/134	120/208	102/116
		(46%)	(90%)	(58%)	(88%)
FU-LSM <25kPa	222 (69%)	114/222	94/102	94/208	108/116
		(51%)	(92%)	(45%)	(93%)
FU-PLT >150G/L	62 (19%)	15/62 (24%)	193/262	193/208	47/116
			(74%)	(93%)	(41%)
FU-LSM <12kPa and FU-PLT	31 (9.6%)	2/31 (6.5%)	206/293	206/208	29/116
>150G/L			(70%)	(99%)	(25%)
FU-LSM <15kPa and FU-PLT	43 (13%)	5/43 (12%)	203/281	203/208	38/116
>150G/L	FO (40%()		(72%)	(98%)	(33%)
FU-LSM <20kPa and FU-PLT	53 (16%)	9/53 (17%)	199/271	199/208	44/116
>150G/L (Baveno VI criteria)			(73%)	(96%)	(38%)

Journal Pre-proof							
FU-albumin>36g/L and FU- PLT>120G/L (RECIST-HCV	<mark>102 (32%)</mark>	<mark>34/102</mark> (33%)	<mark>169/214</mark> (79%)	<mark>169/203</mark> (83%)	<mark>68/113</mark> (60%)		
criteria) ⁴							

¹ Corresponding to the false-negative rate or the reciprocal of the negative predictive value when not meeting the criterion; ² Corresponding to the positive predictive value for FU-CSPH when not meeting the respective criterion; ³ Calculated for patients not meeting the respective criterion; ⁴ Available in 234 cACLD patients and 316 in the overall cohort

Abbreviations:

BL baseline

cACLD compensated advanced chronic liver disease

FU follow-up

LSM liver stiffness measurement

PLT platelet count

ournalpre

Overall cohort: n=675

- Reiberger et al (2012) n=80
- Lens et al (2015) n=100
- Mandorfer and Kozbial et al (2016)/ Schwabl and Mandorfer et al (2018)/ Mandorfer et al (2020) – n=90
- Lens et al (2017)/Lens and Baiges et al (2020) n=226
- Puente et al (2017) n=8
- Mauro et al (2018) n=112
- Abadia et al (2019) n=33
- Diez et al (2020) n=26



n=418 included to study the dynamics of HVPG after HCV-cure (HVPG-cohort)



n=324 included to evaluate the diagnostic performance of NITs (HVPG/NIT cohort)















HIGHLIGHTS

- Pooled analysis on hepatic venous pressure gradient and liver stiffnessmeasurement (LSM)/platelet count (PLT) in advanced chronic liver disease (ACLD) patients achieving HCV-cure
- Post-treatment LSM/PLT can estimate the probability of clinically significant portal hypertension (CSPH) and predict clinical outcomes in compensated ACLD (cACLD)
- cACLD patients with LSM<12kPa & PLT>150G/L (CSPH-excluded; no decompensation risk) may be discharged from portal hypertension surveillance
- cACLD patients with LSM≥25kPa require surveillance/treatment (CSPH-ruledin; increased decompensation risk)